Author's response to reviews

Title: Management of high-risk patients with hypertension and left ventricular hypertrophy in Germany: differences between cardiac specialists in the inpatient and outpatient setting

Authors:

Heinz Voeller (heinz.voeller@klinikamsee.com)
Frank Sonntag (f.j.sonntag@t-online.de)
Joachim Thierry (thiery@medizin.uni-leipzig.de)
Karl Wegscheider (karl.wegscheider@t-online.de)
Friedrich C. Luft (luft@fvk-berlin.de)
Kurt Bestehorn (kurt_bestehorn@msd.de)

Version: 3 Date: 20 September 2006

Author's response to reviews:

Dear Sir,

We have addressed all requested revisions in the attached manuscript.
Title & Abstract - Please ensure these are identical in the manuscript and in the submission system.
Answer: DONE.
Highlighting/Tracking - Please remove any highlighting/tracking from your manuscript text. Answer: DONE.
Methods- Please provide the details of the ethics committee that granted ethical approval for the study.
Answer: DONE (ethics committee of the Physicians Chamber Germany).
Conclusions- Please include a brief Conclusions section after the Discussion.
Competing interests- For those authors who do not have competing interests to declare, the declaration should read 'All other authors declare that they have no competing interests'. Answer: DONE.
The formatting was been adopted according to the Journal's guidelines.
We hope the current version is acceptable to the editors.
With best regards

Heinz Voeller
Kurt Bestehorn